Author's response to reviews

Title: Measuring the tail of the dog that doesn't bark in the night: the case of the national evaluation of Choose Life (The National Strategy and Action Plan to Prevent Suicide in Scotland)

Authors:

Mhairi Mackenzie (m.mackenzie@clinmed.gla.ac.uk)
Avril Blamey (avril.blamey@health.scot.nhs.uk)
Emma Halliday (emma.halliday@health.scot.nhs.uk)
Margaret Maxwell (margaret.maxwell@ed.ac.uk)
Allyson McCollam (allyson@sdcmh.org.uk)
David McDaid (d.mcdaid@lse.ac.uk)
Joanne McLean (joanne@sdcmh.org.uk)
Amy Woodhouse (amy@sdcmh.org.uk)
Stephen Platt (steve.platt@ed.ac.uk)

Version: 4 Date: 5 July 2007

Author's response to reviews:

Dear editorial team - I have changed Prof Platt's name on the submission system, ensured that tables 1&3 are constructed as per recommendations and have renamed section headings within the manuscript. I have, however, retained subheadings in the discussion that I'd strongly recommend were retained if possible for the sake of coherence and readability. If they have to be removed then please do so. Best wishes

Mhairi Mackenzie